Author’s response to reviews

Title: Clinical Characteristics Associated with the Onset of Delirium among Nursing Home Residents

Authors:

Evelyn Cheung (evelyn.cheung@medportal.ca)
Sophiya Benjamin (sophiyabenjamin@gmail.com)
George Heckman (ggheckman@uwaterloo.ca)
Joanne Ho (joanneho@mcmaster.ca)
Linda Lee (lee.linda.lw@gmail.com)
Samir Sinha (Samir.Sinha@sinaihealthsystem.ca)

Version: 3 Date: 18 Jan 2018

Author’s response to reviews:

18 January 2018

Giuseppe Bellelli, M.D.
BMC Geriatrics

Dear Dr. Bellelli,

Re: Manuscript ID BGTC-D-17-00532R1: Clinical Characteristics Associated with the Onset of Delirium among Nursing Home Residents

Thank you very much for your response in regard to the above manuscript. We have now revised the manuscript and respond to the reviewers’ helpful suggestions as follows.
Reviewer 1 (Dr. Kenneth Boockvar):

1. As suggested, we have clarified in our title that the target population is long-term nursing home residents. We have also clarified that in the abstract.

2. As suggested, we have removed the word “appropriateness” in the conclusion of the abstract.

We have made every attempt to keep the additions requested by the reviewers as brief as possible.

We hope these changes will meet with your approval and look forward to your response in due course.

Yours sincerely,

Evelyn Ning Man Cheung, MD, MSc
Andrew P Costa, PhD